

Contents lists available at ScienceDirect

Applied Energy



journal homepage: www.elsevier.com/locate/apenergy

Industrial energy communities: Energy storage investment, grid impact and cost distribution

Kjersti Berg^{a,*}, Sverre Stefanussen Foslie^{a,b}, Hossein Farahmand^a

^a Norwegian University of Science and Technology (NTNU), Trondheim, Norway
^b SINTEF Energy Research, Trondheim, Norway

ARTICLE INFO

Keywords: Energy community Industry flexibility Thermal energy storage Battery energy storage Grid impact

ABSTRACT

Energy communities are a way for end-users to contribute to the green shift, by installing distributed energy resources such as photovoltaic panels. The focus of energy community studies has mainly been on residential and not industrial consumers. Industrial consumers can be an important key in energy communities, especially with regard to reducing peak demand in the grid. In this article, we use real measurements from a transformer station and an industrial consumer in Norway to find the optimal size of energy storage in two cases: whether the industrial consumer invests independently or collaborates with the local urban area as an energy community. We assess the cost savings of the energy community and the advantages for the distribution system operator, in terms of cost reduction and peak import reduction for the energy community. Ultimately, we investigate the equitable distribution of cost savings from joint investments between the industry and the local urban area. Our results show that thermal energy storage is the most favourable storage option, due to lower investment costs than battery energy storage systems. Furthermore, we find that optimising the storage sizes for the whole energy community leads to both cost reduction for the energy community and a reduction in maximum import for the local grid. The costs are reduced by 1.8%, while the maximum import is reduced by 5%, compared to the reference case where there are no energy storages. Moreover, the economic incentive for industrial consumers to join energy communities is substantially influenced by the selected cost saving redistribution method.

1. Introduction

To reach the European climate goals, there is a need for increased electrification and distributed energy resources. This is causing a strain on the distribution grid, imposing challenges to for instance keep voltages within operating limits in areas with a high number of new photovoltaic (PV) installations [1] or avoiding congestions in areas with high electrification from end-users and industrial processes [2]. As several sectors wish to transition from fossil fuels to electricity, the strain on the grid is increasing rapidly [3]. In Norway, both the national and regional grids are already at their capacity limits in many places [4]. This means that many consumers who wish to connect have to wait until grid reinforcements can be made, which may take up to several years [5]. This causes a delay in the green shift, since many of the new applications for grid connections are related to electrification and decarbonisation of fossil-based industrial processes. The load in the Norwegian grid is at its peak during the coldest period in winter [6]. If we are able to reduce the load in these hours, it could provide more capacity for new consumers or enable existing consumers to increase their electricity use, which again can lead to reduced emissions. Industry consumers play an important role here, as they often have consumption with high peaks [7], and changing their consumption patterns can make a difference both in the local grid where they are connected and the higher voltage levels.

Energy communities can be one way of organising and incentivising peak load reduction in the grid. An energy community is a legal entity that is controlled by its members. The members can be individuals, small- and medium-sized enterprises, or local authorities. Energy communities often include distributed energy resources such as PV generation, and shared assets such as energy storage [8]. Energy communities can provide services to the grid, but it is crucial that it is a *local* energy community, meaning that the members are physically located in the same grid area [8]. Energy communities with energy storage can help the grid, by mitigating voltage issues [9] or reduce congestions [10], and might therefore be a valuable collaboration partner for the distribution system operator (DSO), if the energy storage is sized properly and the economic incentives are present.

https://doi.org/10.1016/j.apenergy.2024.123908

Received 16 January 2024; Received in revised form 14 June 2024; Accepted 11 July 2024 Available online 24 July 2024

0306-2619/© 2024 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

^{*} Corresponding author. *E-mail address:* kjersti.berg@sintef.no (K. Berg).

Nomenclature		P_l^{\lim}	Limit of grid tariff, level <i>l</i> [MWh/h]
Indices and sets		$P_t^{\text{steam boiler}}$ $P_t^{\text{steam boiler}}$	Steam boiler electric demand [MWn/n]
i, j	Energy community member	heat pump	[MWN/N]
l	Tariff level	Q_t	MWb/b]
m	Month	Oother	Other thermal demand in hour t [M
t	Time step	\mathcal{Q}_t	Process thermal demand [MWh/h]
L	Set of months	\mathcal{Q}_t	Thermal demand at industry consu
Т	Set of time steps	\mathcal{Q}_{t}	hour t [MWh/h]
W	Set of first hours per week	R ^{B, P2E}	BESS power-to-energy ratio [MW/M
Parameters		R ^{T, P2E}	TES power-to-energy ratio [MW/MV
BESS		R ^{prod}	Grid remuneration from production
η^{DESS}	Efficiency of BESS [–]	TC	Total, annual operational cost for
η^{1LO}	Efficiency of TES [-]	-	community [€]
$\eta^{\text{el. Doller}}$	Efficiency of electric boiler [–]		
$\frac{\gamma^{\text{TES, Sel}}}{\overline{\rho_{\text{RES}}}}$	Self-discharge of TES [%]	Variables	
$\underline{\alpha^{\text{BESS}}}, \underline{\alpha^{\text{BESS}}}$	Max. and min. SOC for BESS [%]	β_t^{impexp}	Binary variable for import and exp
$\underline{\alpha^{\text{TES}}}, \underline{\alpha^{\text{TES}}}$	Max. and min. SOC for TES [%]		hour t
Pel. boiler	Capacity of electric boiler [MWh/h]	$\beta_{l,m}$	Binary variable for deciding grid tari
C_i^{alone}	Member <i>i</i> 's cost when operating alone $[\in]$		for month m and level l
$C^{\text{BESS, inv}}$	Investment cost of BESS [€/MWh]	c ^{grid, peak}	Annual cost of monthly peak grid ta
$C^{\text{BESS, op.}}$	Operational cost of BESS [€/MWh/yr]	e^{BESS}	BESS energy capacity [MWh]
C ^{BESS,ann}	Annualised investment cost for BESS	e^{TES}	TES energy capacity [MWh]
anid a	[€/MWh]	p_t^{400V}	Electricity consumption at 400 V
$C_l^{\text{grid}, p}$	Cost of grid monthly peak, level <i>l</i> [€/MW]	60011	hour t [MWh/h]
$C^{\text{grid, vol}}$	Volumetric grid tariff [€/MWh]	p_t^{690V}	Electricity consumption at 690 V
C_t^{spot}	Spot market price in hour $t \in MWh$	BESS ch	hour t [MWh/h]
Ctax	Grid consumption tax [€/MWh]	$p_t^{\text{DESS,CII}}$	BESS charge in hour t [MWh/h]
$C^{\text{TES, inv}}$	Investment cost of TES [€/MWh]	p_t	BESS discharge in hour t [MWh/h]
$C^{\text{TES, op.}}$	Operational cost of TES [€/MWh/yr]	$p_t^{\text{el. boller}}$	Electric boiler consumption in l
$C^{\text{TES,ann}}$	Annualised investment cost for TES	exp	[MWh/h]
	[€/MWh]	p_t^{imp}	Exported electricity in hour t [MWh
C_i	Distributed, annual operational cost for	p_t^{imp}	Imported electricity in hour t [MWh
_	member $i \in \mathbb{C}$	p_t^{industry}	Electricity consumption of industr
E_i	Yearly energy consumption for member i	may	sumer in hour t [MWh/h]
C	[MWI]	P_m^{max}	Peak electricity demand of more
U	Maximum import/export allowed	.el. boiler	[MW/n/n]
IR	[111 V 11/ 11] Interest rate [_]	$q_t^{\text{on concr}}$	Electric Doller thermal output in
M	Large number [_]	ahot water	[1919911/11] Thermal demand for hot water in
N ^B	Lifetime of BESS [v]	q_t	[MWh/h]
NT	Lifetime of TFS [v]	a ^{TES,ch}	TES charge in hour t [MWh/h]
Pindividual peak	Power neak of member <i>i</i> [MWh/h]	$q_t^{\text{TES,dis}}$	TES discharge in hour t [MWh/h]
<i>i</i> psystem peak	Dower peak of member i during queter	Soc BESS	State of charge of RESS in hour t [M
r_i	peak [MWh/h]	soc ^{TES}	State of charge of TES in hour t [M
Parea	Area load in hour t [MWh/h]	soc _t	
pelectric	Electrical demand at industry consumer in	L	
- t	hour t [MWh/h]		
		generation to cover	their demand [15]. Therefore, it might be

Although the definitions of energy communities [11,12] state that small and medium-sized enterprises can be members, the focus of energy community studies has until now been mainly on households, and industrial consumers are rarely considered [13]. The industrial sector is responsible for over one-third of the energy demand globally [14], and can therefore play a key role in reducing emissions. Some industries have processes with high peak demand and thus have an incentive to install energy storage for peak shaving, especially if they have grid tariffs that are based on peak power. However, it can be challenging for some industries to have enough space to install renewable energy

other	Other electrical demand [MWh/h]
steam boiler	Steam boiler electric demand in hour <i>t</i> [MWh/h]
$Q_t^{\text{heat pump}}$	Heat pump thermal production in hour t [MWh/h]
2 ^{other}	Other thermal demand in hour <i>t</i> [MWh/h]
p_t^{process}	Process thermal demand [MWh/h]
t_t^{thermal}	Thermal demand at industry consumer in hour <i>t</i> [MWh/h]
R ^{B, P2E}	BESS power-to-energy ratio [MW/MWh]
R ^{T, P2E}	TES power-to-energy ratio [MW/MWh]
R ^{prod}	Grid remuneration from production [%]
^r C	Total, annual operational cost for energy community [€]
ariables	
impexp t	Binary variable for import and export in hour t
P1,m	Binary variable for deciding grid tariff level for month <i>m</i> and level <i>l</i>
grid, peak	Annual cost of monthly peak grid tariff [€]
BESS	BESS energy capacity [MWh]
TES	TES energy capacity [MWh]
400V t	Electricity consumption at 400 V bus in hour <i>t</i> [MWh/h]
690V t	Electricity consumption at 690 V bus in hour t [MWh/h]
BESS,ch	BESS charge in hour t [MWh/h]
BESS,dis	BESS discharge in hour t [MWh/h]
el. boiler t	Electric boiler consumption in hour t [MWh/h]
exp	Exported electricity in hour t [MWh/h]
imp	Imported electricity in hour <i>t</i> [MWh/h]
industry t	Electricity consumption of industry con- sumer in hour <i>t</i> [MWh/h]
max m	Peak electricity demand of month <i>m</i> [MWh/h]
el. boiler	Electric boiler thermal output in hour t [MWh/h]
hot water t	Thermal demand for hot water in hour t [MWh/h]
TES,ch	TES charge in hour t [MWh/h]
TES,dis	TES discharge in hour t [MWh/h]
oc_BESS	State of charge of BESS in hour <i>t</i> [MWh]
$oc_t^{'}$ TES	State of charge of TES in hour <i>t</i> [MWh]
•	

5]. Therefore, it might be beneficial for industry consumers to collaborate with other areas that have more space for, i.e., rooftop PV generation.

This article investigates the economic viability of an industrial consumer with low temperature (<100 °C) process heating demands participating in an energy community. It explores the potential symbiotic relationship between an industrial consumer equipped with energy storage and an urban area with distributed generation. We use real measurements from a transformer station and an industrial consumer in Norway to investigate the optimal size of energy storage in two cases: the industrial consumer invests considering only its own load, or it invests considering the local urban area load and generation, as an energy community. Additionally, we evaluate the cost savings achieved

by the energy community in relation to the advantages of the DSO. This assessment includes cost reduction resulting from the energy storage investment and a decrease in peak power import to the community. Finally, we investigate an equitable distribution of cost savings between the industry consumer and the local urban area.

1.1. Related literature

Most studies on energy communities focus on households [16], while only a few involve industrial consumers [13]. Regarding the collaboration between industry and urban areas, several studies investigate "Urban-industrial symbiosis", which is waste management (burning waste and using excess heat for heating the urban area), but very few investigate the potential symbiosis between industrial and urban areas with renewable energy [17,18]. One exception is [17], which looked at an urban-industrial sustainable energy community. The article showed that the industrial sector presents a great opportunity to install renewable energy resources and provide renewable energy to a nearby urban area.

Compared to households, industries have more capital to invest in large central energy storage systems. Also, industry consumers often have multi-energy systems, consisting of for instance both thermal and electrical demand, making it relevant to consider different types of energy storage [19]. As shown in [20], many industries can provide demand response since they already have equipped their facilities with control, measurement and communication infrastructure. The benefit of installing energy storage to enable load shifting is that the industry does not need to change its operation and schedule, which can be difficult due to working hours and manpower. Ref. [21] looked at the sizing of BESS for industrial applications in Germany, for four different industrial profiles. The results showed that it was profitable to install battery energy storage systems (BESS) in most scenarios, due to the peak power grid tariff. In general, however, when comparing BESS and hot water thermal energy storage (TES), studies find that TES is more economical [22,23]. Ref. [23] investigated a hot water TES tailored for an industrial consumer (dairy). The study showed that TES provides cost-effective flexibility, which eliminates the need for BESS, and that the TES contributes to cost reductions and CO₂ emission reduction.

Shared energy storage is considered more economical than individual storage, due to economies of scale and diversity factors [24]. The majority of literature on shared energy storage in energy communities considers households or industry clusters separately. In contrast, this article investigates how energy storage located at an industry consumer can be used in an energy community setting. Concerning shared assets at industrial parks, [25] examined shared energy storage in industrial parks with PV generation. The authors found that shared energy storage increased the local consumption of PV generation. Notably, they did not consider any particular energy storage technology; instead, they used a generic energy storage model in an optimisation model where the objective was to maximise the daily net income from electricity trading. Ref. [26] determined the optimal BESS sharing scheme in an industrial park, by minimising the total operating cost of the industrial park. The authors found that centralised shared energy storage resulted in lower electricity costs and greater utilisation, compared to distributed energy storage at each industry. Energy community studies with energy storage focus mostly on batteries, and only a few works analyse thermal technologies [16], although TES is more costcompetitive than BESS [24]. Ref. [27] looked at peer-to-peer trading in an industrial site in Norway with a community BESS. The objective of the optimisation was to minimise the total cost of electricity for the community. The authors found that peak shaving was the most important factor in cost reduction, due to the monthly peak power grid tariff. Refs. [25-27] provided valuable knowledge on the operation of shared energy storages, however, they did not investigate the optimal size of the storage systems. Ref. [28] investigated the optimal size of BESS for a renewable energy community located in the low-voltage

grid. The authors formulated an optimisation model which minimised investment and operational costs of the energy community. The results showed that the optimal BESS size varied significantly depending on whether the energy community consisted of only residential loads or both residential and commercial loads.

The question of how to distribute the costs and benefits within energy communities is still an open question [16,29,30]. Several studies have investigated the topic [25,31,32]. Ref. [31] examined how a grid company could own an energy storage and dynamically allocate shares to prosumers, while resolving overvoltage and congestion issues. The authors formulated an optimisation model that maximised the total export limit of the residential prosumers. The results showed that community energy storage could reduce active power curtailment and increase total generation. Ref. [32] examined household scenarios and proposed a pricing mechanism for energy storage sharing in a market-oriented environment with several communities. The authors formulated a planning model for community energy systems, where the objective was to minimise the total planning and operating costs for each community manager. The results showed that such a storage sharing mechanism could increase the profit of the communities, reduce the interaction with the grid and increase the self-consumption of renewable energy. In [25], which looked at industrial consumers, the authors proposed a reputation factor to reflect the fairness of energy sharing, where the entity that provided more resources received more rewards. The reputation factor was based on the total amount of electricity shared by the user with the energy storage and increased the net present value of the energy storage.

1.2. Contributions

As the related literature reveals, there is a lack of studies on industry consumers in energy communities, and a lack of energy community studies considering energy storage for other types of energy carriers than electricity (i.e. batteries). Considering these gaps, the contributions of this paper are as follows:

- We investigate the storage investment decision of community electrical and thermal energy storage for an energy community with an industrial consumer and an urban area with distributed generation.
- We provide an optimisation model of an industrial consumer participating in an energy community, by using real, hourly measurements for one year from the industrial consumer and the distribution grid.
- We study the incentives for the industrial consumer to participate in the energy community by assessing equitable methods for distributing costs and benefits stemming from investments in distributed generation and energy storage.

1.3. Outline of paper

The outline of the paper is as follows: Section 2 describes the concept of the paper, the optimisation model formulation and the cost allocation methods. Section 3 describes the real-life case study from Trondheim, Norway and assumptions regarding energy storages. Section 4 presents the results from the optimisation model, cost distribution and sensitivity analysis. Finally, Section 5 gives the concluding remarks.

2. Method

Fig. 1 shows the concept of the paper: An industry consumer is considering an investment in energy storage to minimise its own energy costs. The industrial consumer has the option to invest in a BESS and/or a hot water TES on the thermal side of the system, offering flexibility in electricity consumption while ensuring the seamless operation of



Fig. 1. Paper concept.

the facility's processes. The industry receives its electricity supply from a transformer that also serves the local urban area. The DSO aims to reduce peak imports to the region. We investigate two cases for storage investment: the industry optimises for itself, or the industry and local urban area form a collaborative energy community that optimises together with aggregated net metering. If the energy community is formed, it receives a common grid tariff from the DSO based on the aggregated net metering. The different energy storage technologies are optimised based on this common grid tariff, and the final costs are redistributed to the members by the community manager subsequently.

2.1. Optimisation model formulation

The optimisation model is a deterministic, mixed integer non-linear optimisation program (MINLP) that determines the sizes of BESS and TES. Subsequently, the investment costs, operational costs, storage operation and import/export are outputs of the optimisation model.

The objective of the optimisation model is to minimise annualised capital and operational costs, as given in (1).

$$\min \quad C^{\text{BESS, ann}} e^{\text{BESS}} + C^{\text{TES, ann}} e^{\text{TES}}$$

$$+ C^{\text{BESS, op.}} e^{\text{BESS}} + C^{\text{TES, op.}} e^{\text{TES}}$$

$$+ c^{\text{grid, peak}}$$

$$+ \sum_{t \in T} \left[(C_t^{\text{spot}} + C^{\text{grid, vol}} + C^{\text{tax}}) p_t^{\text{imp}} - C_t^{\text{spot}} (1 + R^{\text{prod}}) p_t^{\text{exp}} \right]$$

$$(1)$$

The different parts of the objective function are as follows. Annualised investment costs for BESS and TES, $C^{\text{BESS},\text{ann}}$ and $C^{\text{TES},\text{ann}}$, and energy storage sizes of BESS and TES, e^{BESS} and e^{TES} . Operational costs for BESS and TES, C^{BESS} , op. and $C^{\text{TES},\text{ op}}$. Annual costs for monthly peak grid tariff, $c^{\text{grid},\text{peak}}$. Annual costs of buying electricity from the spot market, C_t^{spot} , volumetric grid tariff, $C^{\text{grid},\text{ vol}}$, and taxes, C^{tax} , all dependent on the electricity import, p_t^{imp} . Annual revenue for selling electricity to the grid, consisting of the spot market price and compensation from the DSO for reducing losses in the grid, R^{prod} , dependent on the electricity export, p_t^{exp} .

The calculation of the monthly peak grid tariff requires binary variables $\beta_{1,m}$ and $\beta_{2,m}$ to decide the cost level for each month *m* in the set of months *L*, as given in (2). This grid tariff is based on a stepwise cost structure, where the consumer pays depending on which step the monthly peak load falls within.

$$c^{\text{grid,peak}} = \sum_{m \in L} \left[C_1^{\text{grid, } p} p_m^{\max} (1 - \beta_{1,m}) + C_1^{\text{grid, } p} P_1^{\lim} \beta_{1,m} \right]$$
(2)

$$+ C_{2}^{\text{grid, } p}(p_{m}^{\text{max}} - P_{1}^{\text{lim}})(1 - \beta_{2,m})$$

+ $C_{2}^{\text{grid, } p}(P_{2}^{\text{lim}} - P_{1}^{\text{lim}})\beta_{2,m}$
+ $C_{3}^{\text{grid, } p}(p_{m}^{\text{max}} - P_{2}^{\text{lim}})\beta_{2,m}$

Here, $C_1^{\text{grid}, p}$ - $C_3^{\text{grid}, p}$ are the costs of level 1–3 and p_m^{max} is the peak load of month *m*. $\beta_{1,m}$ and $\beta_{2,m}$ take the value of 1 if the monthly peak load is above the limit of P_1^{lim} or P_2^{lim} , respectively.

The annualised investment costs for BESS and TES are calculated as follows:

$$C^{\text{BESS,ann}} = \frac{IR(1+IR)^{N^B}}{(1+IR)^{N^B} - 1} \cdot C^{\text{BESS, inv}}$$
(3)

$$C^{\text{TES},\text{ann}} = \frac{IR(1+IR)^{N^{T}}}{(1+IR)^{N^{T}}-1} \cdot C^{\text{TES},\text{ inv}}$$
(4)

where IR is the investment rate, N^B and N^T are the lifetimes of BESS and TES, and $C^{\text{BESS, inv}}$ and $C^{\text{TES, inv}}$ are the investment costs of BESS and TES, respectively.

Fig. 2 shows a visualisation of a general industrial energy community, including the variable names used in the following equations. The electricity balance constraint for grid exchange is given in (5). The import and export are decided from the industry load, p_t^{industry} , and the load and generation in the urban area, P_t^{area} . When the industry is optimising only for itself, P_t^{area} is set to zero.

$$p_t^{\text{industry}} + P_t^{\text{area}} = p_t^{\text{imp}} - p_t^{\text{exp}} \qquad t \in T$$
(5)

In general, the import to the industry consumer can be described as the electric demand at the industry, P_t^{electric} , the BESS charge/discharge power, $p_t^{\text{BESS, ch}}$ and $p_t^{\text{BESS, dis}}$, and the power consumed by the electric boiler which converts electricity into heat, $p_t^{\text{el. boiler}}$:

$$p_t^{\text{industry}} = P_t^{\text{electric}} + p_t^{\text{BESS, ch}} - p_t^{\text{BESS, dis}} + p_t^{\text{el. boiler}} \qquad t \in T \quad (6)$$

The electric boiler converts electricity to heat, $q_t^{\text{el. boiler}}$, with an efficiency, $\eta^{\text{el. boiler}}$. This heat is then connected to the thermal demand, Q_t^{thermal} and the TES charge/discharge, $q_t^{\text{TES, ch}}$ and $q_t^{\text{TES, dis}}$:

$$p_t^{\text{el. boiler}} \eta^{\text{el. boiler}} = q_t^{\text{el. boiler}} \qquad t \in T$$
(7)

$$q_t^{\text{el. boiler}} = Q_t^{\text{thermal}} + q_t^{\text{TES, ch}} - q_t^{\text{TES, dis}} \qquad t \in T$$
(8)



Fig. 2. General overview of the energy flows in the energy community. The urban area is modelled as a parameter representing load and generation, while the industry consumer consists of an electric and a thermal part, where BESS and TES can be invested in

To avoid import and export at the same time, a binary variable, β_t^{impexp} , is added in the following restrictions, where *G* is the maximum limit for import and export from/to the grid:

$$p_{t}^{\text{imp}} \leq \beta_{t}^{\text{impexp}} \cdot G \qquad \qquad t \in T \tag{9}$$

$$p_t^{\exp} \le (1 - \beta_t^{\operatorname{impexp}}) \cdot G \qquad t \in T(10)$$

Eq. (11) keeps track of the highest hourly consumption of month *m*, while (12)–(15) use binary variables $\beta_{1,m}$ and $\beta_{2,m}$ to determine which level the peak power falls within for month m, using the large number M.

$$p_t^{\min} \le p_m^{\max} \qquad t \in T$$
(11)

$$p_m^{\max} - P_1^{\lim} \le \beta_{1,m} M \qquad \qquad m \in L$$
 (12)

 $P_1^{\lim} - p_m^{\max} \le (1 - \beta_{1,m})M$ $m \in L$ (13)

$$p_m^{\max} - P_2^{\lim} \le \beta_{2,m} M \qquad \qquad m \in L$$
 (14)

$$P_2^{\lim} - p_m^{\max} \le (1 - \beta_{2,m})M \qquad m \in L$$
(15)

The industry consumer has the possibility to invest in a BESS and/or a hot water TES. The constraints for the BESS are given in (16)-(22), where η^{BESS} denotes the charge/discharge efficiency of the BESS. Eq. (20) ensures that the state of charge (SOC) of the battery, soc_t^{BESS} is the same at the beginning of every week, to avoid long-term scheduling of the battery system in the optimisation model. This constraint is therefore only valid for hours in set W, which is the set of the first hour of each week. The SOC is restricted to maximum and minimum levels, α^{BESS} and $\overline{\alpha^{\text{BESS}}}$, to limit degradation in (18)–(19). The discharge and charge variables are further restricted by the energy capacity times the power-to-energy ratio, $R^{B, P2E}$, representing the inverter size of the battery system.

$$soc_{t}^{\text{BESS}} = soc_{t-1}^{\text{BESS}} + \eta^{\text{BESS}} p_{t}^{\text{BESS, ch}} \Delta t \qquad t \in T \setminus \{1\}$$
(16)
$$- \frac{1}{\text{BESS}} p_{t}^{\text{BESS,dis}} \Delta t$$

$$soc_t^{\text{BESS}} = soc_t^{\text{BESS}} + \eta^{\text{BESS}} p_t^{\text{BESS, ch}} \Delta t \qquad t = 1 \quad (17)$$
$$- \frac{1}{n^{\text{BESS}}} p_t^{\text{BESS, dis}} \Delta t$$

 $t \in T$ (18)

$$soc_t^{\text{BESS}} \le \overline{\alpha^{\text{BESS}}}e^{\text{BESS}}$$

 $t \in T$ (18)
 $soc_t^{\text{BESS}} \ge \alpha^{\text{BESS}}e^{\text{BESS}}$
 $t \in T$ (19)

$$soc^{\text{BESS}} = soc^{\text{BESS}}$$
 $t \in W$ (20)

$$\frac{1}{3^{\text{BSS}}}, \text{ th} < R^{\text{B}}, P^{\text{E}}e^{\text{BESS}} \qquad t \in T \quad (21)$$

 $S, dis \leq R^{B, P2E} e^{BESS}$ $t \in T$ (22)

The TES is modelled similar to the BESS, except a self-discharge parameter, $\gamma^{\text{TES},sel}$, which is added to represent heat leakage:

$$soc_{t}^{\text{TES}} = soc_{t-1}^{\text{TES}} + \eta^{\text{TES}} q_{t}^{\text{TES}}, \text{ ch} \Delta t \qquad t \in T \setminus \{1\}$$
(23)
$$- \frac{1}{n^{\text{TES}}} q_{t}^{\text{TES}, \text{ dis}} \Delta t - soc_{t-1}^{\text{TES}} \gamma^{\text{TES}, \text{ sel}}$$

$$soc_{t}^{\text{TES}} = soc_{T}^{\text{TES}} + \eta^{\text{TES}} q_{t}^{\text{TES}, \text{ ch}} \Delta t \qquad t = 1 \quad (24)$$
$$- \frac{1}{n^{\text{TES}}} q_{t}^{\text{TES}, \text{ dis}} \Delta t - soc_{T}^{\text{TES}} \gamma^{\text{TES}, \text{ sel}}$$

$$soc_t^{\text{TES}} \le \overline{\alpha^{\text{TES}}}e^{\text{TES}}$$
 $t \in T$ (25)

$$soc_t^{\text{TES}} \ge \underline{\alpha^{\text{TES}}}e^{\text{TES}}$$
 $t \in T$ (26)

$$q_t^{\text{TES, ch}} \le R^{\text{T, P2E}} e^{\text{TES}} \qquad t \in T$$
(27)

$$q_t^{\text{TES, dis}} \le R^{\text{T, P2E}} e^{\text{TES}} \qquad t \in T$$
(28)

Here, soc_t^{TES} denotes the SOC of the TES, η^{TES} is the charge/ discharge efficiency, α^{TES} and $\overline{\alpha^{\text{TES}}}$ are the minimum and maximum SOC limits, and $R^{T, P2E}$ is the power-to-energy ratio of the TES.

Eqs. (29) and (30) ensure that variables are non-negative. Eqs. (31)-(32) define the binary variables.

$$\begin{aligned} e^{\text{BESS}}, e^{\text{TES}}, p_t^{\text{exp}}, p_t^{\text{imp}}, p_t^{\text{industry}}, p_t^{\text{BESS, ch}}, \\ p_t^{\text{BESS, dis}}, soc_t^{\text{BESS}}, q_t^{\text{TES,ch}}, q_t^{\text{TES,dis}}, soc_t^{\text{TES}}, \\ q_t^{\text{TES, sel}}, p_t^{\text{el. boiler}}, q_t^{\text{el. boiler}} \ge 0 \qquad t \in T \quad (29) \end{aligned}$$

$$p_m^{\max} \ge 0$$
 $m \in L$ (30)

$$\beta_{1,m}, \beta_{2,m}, \beta_{2,m} \in \{0,1\}$$
 $m \in L$ (31)

$$\beta_t^{\text{impexp}} \in \{0, 1\} \qquad t \in T \quad (32)$$

2.2. Cost distribution methods

After running the optimisation model, we compare different ways of distributing the operational costs to the industry consumer and the urban area.

The first method is the flat energy pricing [29], where the cost for each member i, C_i , is calculated from the total costs, TC, and the yearly energy consumption of each member i, E_i :

$$C_i = \frac{TC}{\sum_i E_i} \cdot E_i \tag{33}$$

The second method is the coincident peak [29], where the cost is calculated on the basis of each member's power consumption in the hour of the system peak, $P_i^{\text{system peak}}$:

$$C_{i} = \frac{TC}{\sum_{i} P_{i}^{\text{system peak}} \cdot P_{i}^{\text{system peak}}}$$
(34)

The third method is the non-coincident peak [29], where the cost is calculated similarly to the previous method, but the individual peak of each member, $P_i^{\text{individual peak}}$, is used instead of the power consumption at the system hour:

$$C_{i} = \frac{TC}{\sum_{i} P_{i}^{\text{individual peak}}} \cdot P_{i}^{\text{individual peak}}$$
(35)

The fourth method is based on the Shapley value [30], where the costs are distributed among members depending on their contribution to the total cost. The Shapley value is a concept from cooperative game theory, where the payoffs are distributed among players in a coalitional game [33]. When there are two actors in the energy community, the cost is calculated as follows for member i:

$$C_i = \frac{(TC - C_i^{\text{alone}}) + C_j^{\text{alone}}}{2}$$
(36)

where C_i^{alone} is member *i*'s cost when operating alone, hence before joining the energy community. C_j^{alone} is the cost before joining the energy community for the other actor, *j*.

3. Case study

The presented model is applied to a case study in Trondheim, Norway. The industry consumer is a real-life dairy producing around 75 million litres of dairy products yearly. Processing of dairy products includes both heating and cooling demands, for processes such as pasteurising, cleaning, storage, and product cooling etc. [34]. We focus on the process heating demands of the investigated dairy in this work, which require hot water slightly below 100 °C. The dairy therefore requires both thermal and electrical energy and is connected to a transformer that also supplies an urban area of diverse consumers. The dairy and the DSO of the area have provided hourly measurements for one year, 2022. The available space for accommodating energy storage solutions is a critical consideration for industrial consumers. In our case study, the industrial site is able to fit a large hot water tank and/or a battery container, and there are therefore no size limitations on the energy storages.

More details on the specific constraints for the industry consumer can be found in Appendix A. All measurements from the industry consumer are from 2022, and specifications regarding the data handling can be found in Appendix B.

3.1. Load and generation

Fig. 3 shows the annual load of the industry consumer (dairy), and the urban area load with and without PV generation. The urban area load is taken from measurements at the transformer station in 2022, provided by the DSO in the area, where the industry load is subtracted. The increase in load at the end of the year is attributed to the decrease in temperatures in December. Figs. 4 and 5 show the yearly consumption and the number of consumers connected to the transformer station. Note that "Industry" in these figures denotes the total amount of industry consumers in the area, not only the dairy. The majority of the load comes from households, both in terms of number of customers and their yearly consumption (35%). Commercial buildings and services also contribute significantly to the yearly consumption (31%), although there are only 223 of them, compared to 2452 households. 33 industries stand for 31% of the yearly consumption in the area, and this includes the industrial consumer (dairy).

In this case study, we assume that the urban area has installed PV generation. The PV is sized based on the urban area load shown in Fig. 5. There are 2452 households, and the average PV size in Norway is 9 kWp for households [35]. We assume that 50% of the households install PV panels, which amounts to $2452 \cdot 9 \cdot 50\%$ kWp ≈ 11 MWp. This PV generation could in theory be located anywhere in the urban area, as the load profile is aggregated. The PV profile is generated from Renewables Ninja [36] for the location of Trondheim, Norway.

The characteristics of the industry consumer (dairy) load and urban area load are presented in Table 1. The term "Area" represents the urban area load data measured at the transformer (with the industry load subtracted), while "Area w/ PV" denotes the urban area load when the synthetic PV generation is subtracted. The coincidence factor¹ of



Fig. 3. Annual load and PV generation for industry consumer and urban area for year 2022. Net demand is positive for load and negative for generation.



Fig. 4. Yearly consumption at transformer station divided into customer group. "Industry" denotes the total amount of industry consumers in the area, not only the dairy.



Fig. 5. Number of consumers at transformer station divided into customer groups. "Industry" denotes the total amount of industry consumers in the area, not only the dairy.

Table 1

Load characteristics. "Industry" refers to the dairy.

	Industry	Area	Area w/PV
Peak load [MWh/h]	4.00	14.94	14.94
Annual load [MWh]	17,241	59,729	49,738

the urban area and the industry consumer is 0.93, and does not change when we add the PV generation, since the PV generation is zero at the peak load hour. We see that the dairy consumes 22% of the annual load in total. When we add PV generation to the urban area load, the annual load is reduced by 9991 MWh.

 $^{^1\,}$ Coincidence factor is calculated as the maximum total load divided by the sum of individual maximum load: 17.69/(4.00 + 14.94).

Table 2

Monthly peak grid tariff [37].

p_m^{max}	Winter [€/kW]	Summer [€/kW]
<100	4.6	3.0
>100 and <500	3.8	2.6
>500	3.0	21



Fig. 6. Spot price for the year 2019 and volumetric grid tariff, including consumption tax and VAT.

3.2. Grid tariff and spot price

The industry consumer has a volumetric grid tariff and a monthly peak grid tariff. The volumetric tariff is 0.39 ¢/kWh, and the consumption tax is 0.055 ¢/kWh [37]. The monthly peak grid tariff for the industry consumer is shown in Table 2 for winter (November-April) and summer (May-October) [37]. The industry pays for each step of the tariff. For instance, if the monthly peak consumption in January is 650 kWh/h, the cost would be $100 \cdot 4.6 + 400 \cdot 3.8 + 150 \cdot 3.0 \in = 2430 \in$.

Fig. 6 shows the spot price from 2019 used as input, both with and without the volumetric grid tariff and consumption tax (excluding the monthly peak grid tariff). The spot prices are skewed to match the weekdays of 2022.

3.3. Thermal energy storage

The TES is assumed to be a hot water tank TES, with the hot water outlet near the top of the tank, and the cold water return near the bottom of the tank. In an idealised storage, perfect thermal stratification can be assumed, meaning that there is no mixing of hot and cold water within the tank, and the temperature increases from the bottom to the top [38]. To account for imperfections in the stratification, as well as the requirement for a sufficiently high temperature difference between the TES outlet and the process demands, we assume that the upper and lower SOC limits of the thermal storage are 75% and 25%, respectively. The cost of a TES is highly subjected to economies of scale [39], but in the capacity range pertinent to this study, we assume a linear cost per energy unit of 25 /kWh [40].

3.4. Battery energy storage system

The BESS is assumed to be a lithium-ion battery, and is modelled with an investment cost dependent on battery energy capacity of 200 ϵ/kWh [41]. According to [41], this investment cost is assumed to be reached within 2030 for nickel manganese cobalt oxide (NMC), lithium manganese oxide (LMO) and nickel cobalt aluminium (NCA) technologies. Furthermore, the battery SOC is restricted to be between 10% and 90%.

Applied Energy 373 (2024) 123908

Table 3	
---------	--

nput parameters for optimisation.	
Parameter	Value
Investment cost BESS, C ^{BESS, inv}	200 €/kWh [41]
OPEX BESS, C ^{BESS, op.}	0.57 €/MW/yr [42]
Investment cost TES, CTES, inv	25 €/kWh [40]
OPEX TES, C ^{TES, op.}	8.8 €/MWh [42]
Efficiency BESS, η^{BESS}	0.98 [41]
Efficiency TES, η^{TES}	0.9 [40]
Lifetime BESS, N ^B	15 y [41]
Lifetime TES, N^{T}	20 y [40]
TES discharge, $\gamma^{\text{TES,sel}}$	0.2%
Interest rate, IR	0.051
Capacity electric boiler, Pelectric boiler	2500 kW
Efficiency electric boiler, $\eta^{\text{electric boiler}}$	0.99
Power-to-energy ratio BESS, R ^{B, P2E}	1
Power-to-energy ratio TES, $R^{T, P2E}$	1
BESS SOC upper and lower limits	90%, 10%
TES SOC upper and lower limits	75%, 25%

3.5. Technical input parameters to optimisation model

Table 3 shows the input parameters for the optimisation model.

4. Results and discussion

This section shows the results from running the optimisation model for the different cases. We investigate three cases:

- Reference (Ref.): No storage investment.
- Industry: We investigate the sizing of energy storage when the industry consumer optimises its own investment and operational costs without considering the urban area. The optimisation model exclusively considers the industry consumer's energy demand.
- Energy Community (EC): We investigate the sizing of energy storage when the industry consumer collaborates with the area. The optimisation model is executed to account for both the industrial and urban energy demand, alongside PV generation.

The model is formulated as a MINLP, and is implemented in Pyomo/Python [43], using Gurobi [44] as an optimisation solver. The model was run for hourly values of one year, which took approximately 84 s on a Dell PC with an Intel Core i5-1145G7 @ 2.60 GHz, 16 GB RAM.

First, we display the results from the optimisation model in two reference weeks. Next, we compare the annual costs and maximum import for the cases. After this, we investigate different methods for distributing the costs between the industrial consumer and the urban area. Finally, a sensitivity analysis is carried out to investigate the impact of several input parameters.

4.1. Summer and winter reference weeks

Two reference weeks for winter and summer are based on the maximum transformer load (12–19 December) and minimum transformer load (01–08 August). Here, the results are presented for these reference weeks for Case Industry and Case Energy community.

4.1.1. Case industry

Fig. 7 shows the import to the industry consumer for the winter week, divided into the two buses where the energy storages can be placed, BESS can be placed at the electrical bus (blue) and TES can be installed at the thermal bus (green). The optimal TES size is found to be 4.69 MWh, and the model does not invest in a BESS. The original import to the industry is shown in a dashed line, and it is clear that the TES is operating to reduce the peak demand, due to the monthly peak grid tariff. In this week, the peak is reduced from 3.87 MWh/h to 3.23 MWh/h.



Fig. 7. Import to industry consumer in Case Industry, winter week. The industrial consumer has both electrical and thermal consumption, and the difference between the original and the new import is due to the TES operation.



Fig. 8. Import to energy community in Case Industry, winter week. The import to the energy community is divided into the electrical and thermal demand of the industrial consumer, and the load and PV generation of the urban area. The difference between the ref. import and the new import is due to the TES operation.

Fig. 8 shows the import to the energy community for the winter week. The total import is due to electrical and thermal load at the industry consumer, and load and PV generation of the urban area. At hour 8290, the peak load is 17.15 MWh/h, where the urban area (purple) has the highest share of the load, with 13.82 MWh/h. The change between the original import (stippled line) and the import when storage is included (black line) is due to the change in operation from the electric boiler together with the TES. Although the objective in this case is to reduce the peak load for the industry, not the whole urban area, the import to the energy community is in fact reduced from 17.69 MWh/h to 17.15 MWh/h, a reduction of 3.1%.

4.1.2. Case energy community

When the optimisation model is run for Case EC, which accounts for both the industry load and the urban area load and PV generation, the model invests in a larger TES of 5.46 MWh. BESS is still not profitable to invest in. Fig. 9 compares the import to the energy community for the winter week for all cases. It can be seen that the maximum import, which originally occurred at hour 8290, is lowered even further in Case EC, compared to Case Industry, to 16.78 MWh/h.

In the summer, the PV generation of the urban area affects the import/export a lot more than in winter, and the load is much lower



Fig. 9. Import to energy community for all cases, winter week. Ref. (no PV) represents the load in the reference case without PV generation, and is included to illustrate the low effect of PV generation in winter. The peak import is reduced in Case Industry and Case EC due to the TES operation.



Fig. 10. Import (positive) and export (negative) to/from energy community in Case EC, summer week. The import/export is divided into the electrical and thermal demand of the industrial consumer, and the load and PV generation of the urban area. The peak import is reduced compared to the reference case, due to the TES operation.

than in winter. Fig. 10 shows the import to the energy community for Case EC in the summer week. As in Fig. 8, the import consists of the industry electrical and thermal load, and the urban area load and PV generation. The peak demand in this week originally occurred at hour 5100 with 9.98 MWh/h, without PV generation. The highest peak in the summer week now occurs in hour 5178, and is 8.35 MWh/h. Interestingly, the TES is only operating in the last two days of the summer week, as shown in Fig. 11. The figure compares the TES operation in Case EC and Case Industry for the summer week, and it is clear that the TES charges and discharges more often in Case Industry. In fact, the annual TES use in Case Industry is 2818 h, while in Case EC it is 2224 h.

When comparing the import to the energy community for all three cases for the summer week, we see that PV has a large significance, as shown in Fig. 12, creating valleys (low demand) at new times of the day. The valleys now occur around noon, instead of during the night/early morning, which originally happened in Case Ref. (no PV).

Fig. 13 shows a violin plot of the import/export to/from the energy community per month for all cases. The figure illustrates a substantial



Fig. 11. TES charge (positive) and discharge (negative) for Case Industry and Case EC, summer week.



Fig. 12. Import (positive) and export (negative) to/from energy community for all cases, summer week. Ref. (no PV) represents the load in the reference case without PV generation, and is included to demonstrate the effect of PV generation in summer.

occurrence of hours with high demand in December, and import fluctuations throughout the year. January and February are also months with high peak demands. It is observed that the peak demand during December and January exhibit a decrease in Case Industry, and to a greater extent in Case EC, compared to the reference case. Case Industry yields a lower peak demand compared to Case Ref. in nine out of twelve months. Consequently, the entire area accrues benefits from the independent optimisation efforts of the industrial consumer. The peak load is, however, always reduced even further when the whole energy community optimises together. It can also be seen that there is an export indicated by negative values from April to October, which never exceeds 5 MWh/h. The storage is only used to reduce peak demand and never used to reduce export in the summer months.

4.2. Comparing costs and maximum import

Table 4 shows a summary of the storage investment and maximum import for the different cases. As already mentioned, there is no BESS investment in any of the cases. When comparing Case Industry and Case EC, the size of the TES is increased by 770 kWh.

The DSO is interested in whether the maximum import to the whole area, and therefore the stress on the transformer, is lowered. For Case Industry, we can see that the maximum import to the area is reduced by 3.1%, from 17.69 MWh/h to 17.14 MWh/h. For Case EC, the maximum import to the urban area is reduced even further, with 5.1%, to 16.78 MWh/h. Hence, optimising for the whole urban area reduces the peak

Table 4

Yearly results - technical

	Case		
	Ref.	Industry	EC
Size BESS [MWh]	0	0	0
Size TES [MWh]	0	4.69	5.46
Max. imp. industry [MWh/h]	4.00	3.34	4.22
Max. imp. EC [MWh/h]	17.69	17.14	16.78

Tol	hle	5

Yearly results - costs [€].

Cost	Case			
	Ref.	Industry	EC	
Ann. inv. cost BESS [k€]	0	0	0	
Ann. inv. cost TES [k€]	0	9.48	11.04	
Op. cost BESS [k€]	0	0	0	
Op. cost TES [k€]	0	0.041	0.048	
Op. cost industry [k€]	1030	999	?	
Op. cost urban area [k€]	3009	3009	?	
Total op. cost [k€]	4039	4007	3968	

load more than optimising for the Industry alone, which is expected due to the increased investment in energy storage. At the same time, it is noteworthy that optimising for the industry consumer alone reduced the peak import to the energy community significantly. Note also that the maximum import to the industry consumer increases in Case EC, compared to both Case Ref. and Case Industry, due to charging of energy storage, which is more beneficial for the peak load of the energy community as a whole.

Table 5 shows the costs for each case, split into annualised investment costs for the energy storage technologies and operational costs for the energy storages, the industry consumer and the urban area. We see that the total operational costs are reduced by 0.8% in Case Industry, and 1.8% in Case EC. For the industry consumer, the operational costs are reduced by 3.0% in Case Industry.

It is intuitive that an energy community with aggregated net metering will experience a cost reduction, compared to the case where it does not have aggregated metering. For the DSO to allow this, the question remaining is whether this gives a benefit to the grid, in terms of a reduction of peak import. This case study shows that both peak import and costs in fact are reduced. Hence, both the energy community and the DSO are better off in this case.

4.3. Equitable distribution of costs within EC

In this energy community, the industry consumer is contributing with energy storage, while the urban area is contributing with PV generation. The total reference operational costs were 4039 k \in , as shown in Table 5, and have been lowered to 3968 k \in from the aggregated net metering. As described in Section 2.2, we investigate four ways of distributing the costs: flat energy pricing, coincident peak, non-coincident peak and Shapley value.

In the flat energy pricing method, the cost is divided based on yearly consumption. In Case EC, the industry consumer has a yearly consumption of 17,642 MWh and the urban area has a yearly consumption of 49,738 MWh:

$$C_I = \frac{3968}{17,642+49,738} \cdot 17,644 = 1039 \tag{37}$$

$$C_A = \frac{3968}{17,642+49,738} \cdot 49,738 = 2929 \tag{38}$$

We see that the industry cost, C_I , is 9 k€ higher than in the reference case. The urban area cost is lower than in the reference case.

In the coincident peak method, the costs are divided depending on how much each member contributes to the system peak. The system



Fig. 13. Violin plot of import (positive) and export (negative) to/from energy community for all cases across different months.

peak occurs in hour 8223 for Case EC, where the industry consumer has a consumption of 2.17 MWh/h and the urban area has a consumption of 14.61 MWh/h:

$$C_I = \frac{3968}{(2.17 + 14.61)} \cdot 2.17 = 513 \tag{39}$$

$$C_A = \frac{3968}{(2.17 + 14.61)} \cdot 14.61 = 3454 \tag{40}$$

Here, the industry cost is 517 kC lower than in the reference case, and also lower than the cost obtained in Case Industry. The urban area cost is higher than in the reference case.

In the non-coincident peak method, the costs are divided depending on each member's individual peak. In Case EC, the peak of the industry consumer is 5.29 MWh/h, while the peak of the urban area is 14.94 MWh/h:

$$C_I = \frac{3968}{(5.29 + 14.94)} \cdot 5.29 = 1037 \tag{41}$$

$$C_A = \frac{3968}{(5.29 + 14.94)} \cdot 14.94 = 2931 \tag{42}$$

Here, the industry operational cost is 7 k \in higher than in the reference case. The urban area cost is lower than in the reference case.

With the Shapley value, the costs are divided depending on each member's contribution to the cost reduction. In this case, we only have two actors since the urban area is aggregated into one:

$$C_I = \frac{(3968 - 3009) + 999}{2} = 979 \tag{43}$$

$$C_A = \frac{(3968 - 999) + 3009}{2} = 2989 \tag{44}$$

Here, both actors save 20 k \in by joining the energy community. Calculating the cost reduction for each member in the urban area is out of the scope of this paper, since we only have aggregated data. One might argue that it is unfair that the industry receives half of the cost reduction, since the urban area consists of many members. At the same time, it is the industry that is investing in energy storage that makes it possible to reduce the peak load in such a manner that the costs are reduced.

Comparing the different cost distribution methods, it is only the Shapley value method that reduces costs for both the industrial consumer and the urban area. In Case Industry, the total costs for the industry are operating costs of 999 k \in and annual investment costs

of TES of 9.48 k \in , a total cost of 1008 k \in . If the industry consumer joins the energy community, and invests in a larger storage, its costs increase if flat energy pricing or non-coincident peak is used as a cost distribution method. If the cost distribution method is the coincident peak, the industry is rewarded for lowering its individual peak (using storage) when the system peak is occurring, and it, therefore, reduces its costs. The downside of using the coincident peak is that the urban area experiences increased costs compared to the reference case, since it is not rewarded for what the storages are doing as they are located at the industry consumer. Therefore, it seems like the Shapley value is the fairest cost distribution method. The drawback of this method is that the cost for the industry consumer and the urban area if they did not join the energy community, needs to be known, while the other methods only rely on smart meter data.

4.4. Sensitivity analysis

Two types of sensitivity analysis have been carried out: first, we stepwise change the investment costs of TES and BESS, to see how this impacts the storage sizes, total costs and maximum import from the grid. Second, the levels of spot price, grid tariff, interest rate and PV size are changed one-at-a-time, to see how this impacts the main results.

4.4.1. Sensitivity of storage capital expenditures

Figs. 14 and 15 show the results of the sensitivity analysis on TES and BESS capital expenditures, for Case Industry and Case EC, respectively. The heatmaps show the stepwise results of running the optimisation model 6×7 times, with varying investment costs for BESS and TES. The value ranges are equal in both figures, so the colours can be compared.

Fig. 14(a) and (b) show the sizes of the BESS and TES, (c) shows the reduction in total costs of the industry consumer, and (d) shows the reduction in maximum import to the energy community. The capital expenditure for BESS is in the range of 100–200 €/kWh, and for TES it is in the range of 10–100 €/kWh. The white colour in the plot indicates no storage investment. It becomes evident that the model only invests in a BESS when the BESS investment cost is 160 €/kWh or lower, and the TES cost is 55 €/kWh or higher. For these cases, the BESS size ranges from 402 kWh to 1.3 MWh. The model invests in a TES for all costs, except when the TES investment cost is above 70 €/kWh



Fig. 14. Heatmap of (a) BESS size, (b) TES size, (c) reduction in total costs for the industry consumer and (d) reduction in maximum import to energy community as the function of TES and BESS investment costs for Case Industry. White colour means no storage investment. A positive reduction in costs means lower costs than the original result. A positive reduction in maximum import means a lower maximum import than the original result.



____ [___,,...]

Fig. 15. Heatmap of (a) BESS size, (b) TES size, (c) reduction in total costs for the energy community and (d) reduction in maximum import to energy community as the function of TES and BESS investment costs for Case EC. White colour means no storage investment. A positive reduction in costs means lower costs than the original result. A positive reduction in maximum import means a lower maximum import than the original result.

and the BESS investment cost is 160 €/kWh or lower. The TESS size ranges from 0.6 MWh to 8.0 MWh, which is quite a large spread. The total cost for the industry consumer and the maximum import, on the other hand, vary very little. The reduction in cost ranges from -1.2% to 0.8%, indicating that the results are robust. The maximum import does not change significantly from the original results, with a reduction in the range of -2.4% to 0.7%. We see that the highest reduction in cost occurs when we have the largest investment in TES. Similarly, the reduction of maximum import follows the increasing investment in TES capacity (note that a negative reduction of maximum import corresponds to an increase in import).

Fig. 15 shows the same heatmap, but for Case EC. Comparing with Fig. 14, we see that there is more investment in BESS for lower TES investment costs. When the TES investment cost is 10 €/kWh, the model invests in a TES with a capacity of 8.2 MWh. Additionally, when BESS investment costs are also low, it is profitable to install a small BESS to reduce the peak import. However, when the TES investment cost rises to 25 €/kWh, the optimal TES size decreases to 5.6 MWh. In this scenario, the model does not find it profitable to combine TES with a BESS for peak import reduction when BESS investment costs exceed 100 €/kWh. Overall, the TES sizes range from 0.4 MWh to 8.2 MWh, and the BESS sizes range from 70 kWh to 1.6 MWh, both larger ranges than for Case Industry. The reduction in total cost ranges from -0.4%to 0.2%, hence a very little change. Maximum import varies more than it did in Case Industry. The reduction is now in the range of -4.2% to 0.7%. Hence, high investment costs for TES lead to smaller TES sizes and therefore an increase in maximum import.

To summarise, we see that the results for total costs are quite robust, as the sensitivity analysis shows only a slight difference in values. There is a larger spread in the storage investment decision, and overall we see that lower storage cost leads to larger storage sizes, lower total costs, and lower maximum import.

4.4.2. Sensitivity of spot price, grid tariff, interest rate and PV size

Table 6 shows the sensitivity analysis of spot price level, grid tariff level, interest rate and PV size for Case EC. The spot price level and grid tariff level are given as factors that are multiplied by the original input. Both the volumetric and the monthly peak grid tariff are adjusted. The interest rate and PV size are given in [%] and [MWp], respectively. Original inputs are given in bold. The overall picture is that the results are quite robust, since they do not change significantly for the different inputs. Two exceptions can be seen: the operational cost is clearly sensitive to the spot price level, and the optimal TES size is clearly sensitive to the grid tariff level.

The TES size increases when the grid tariff level increases, and decreases when the interest rate increases. BESS is only invested in when spot price levels and grid tariff levels are high. Overall, The TES size is in all cases a minimum of 16 times larger than the BESS.

For varying spot price levels, the TES size only varies slightly, ranging from 5.30 to 5.51 MWh. Interestingly, the TES size does not follow the spot price level linearly. For a spot price level of 2, the model invests in a slightly larger storage to be able to reduce the monthly peaks, and thereby the monthly peak grid tariff cost. When the spot price level further increases, however, it outcompetes the price signal from the monthly peak grid tariff, and the model has a higher priority of responding to high/low spot prices, which requires a smaller TES. When the spot price increases to even higher levels, and thereby higher fluctuations, the model finds it profitable to invest in more storage that can be used to lower the cost from importing electricity, and this storage is also used to reduce the monthly peaks. In other words, there is a constant trade-off between the different costs in the objective function: the energy storage investment costs, the spot price and energy grid tariff costs, the monthly peak grid tariff costs and the remuneration from feed-in. The monthly peak grid tariff price signal is dominant until a certain point, but when the spot price gets higher, these two

Table 6

Sensitivity analysis on the spot price level, grid tariff level, interest rate and PV size for Case EC. Spot price level and grid tariff level are multiplication factors of the original input. Numbers in bold correspond to the original assumption.

		e^{BESS}	e^{TES}	Op. costs	Max. imp.
		[MWh]	[MWh]	[k€]	[MWh/h]
	1	0	5.46	3 968	16.78
	2	0.02	5.51	7 277	16.77
Spot price level	3	0	5.30	10,587	16.78
	4	0.07	5.30	13,894	16.72
	5	0.30	5.51	17,198	16.63
	1	0	5.46	3 968	16.78
	1.5	0.05	6.22	4 265	16.74
Grid tariff level	2	0.19	7.38	4 565	16.68
	2.5	0.29	8.22	4 866	16.64
	3	0.59	10.21	5 159	16.52
	3	0	5.73	3 968	16.78
	4	0	5.54	3 968	16.78
Interest rate [%]	5.1	0	5.46	3 968	16.78
	6	0	5.27	3 968	16.78
	7	0	5.06	3 969	16.78
	11	0	5.46	3 968	16.78
	14	0	5.00	3 831	16.78
PV size [MWp]	17	0	5.52	3 693	16.78
	20	0	4.86	3 558	16.78
	23	0	4.80	3 424	16.78

price-signals become conflicting as the optimisation model needs to decide what the optimal monthly peak is, which energy storage size is required to reach this peak, and whether it is more important to charge/discharge the TES when the prices are low/high.

The TES size generally decreases for higher PV sizes, except for a PV size of 17 MWp where the model finds it profitable to invest in a slightly larger TES (5.52 MWh) to reduce the monthly peak power grid tariff. When the PV size increases further, however, the increased generation leads to higher cost savings from electricity import, remuneration from feed-in, and monthly peak power grid tariff during the summer months, and therefore the TES size is reduced.

The total operating cost is highly sensitive to the spot price level, increasing to over 17 M \in for a spot price level increase of 5. The grid tariff level also increases the overall costs. For a change in interest rate, the cost remains in the same range as before, as it mainly impacts the investment decision in energy storage.

The lowest maximum import obtained is 16.52 MWh/h, which is a reduction of 1.5% from the original optimisation result. This result is obtained for the highest grid tariff level. In general, we can see that the maximum import follows the grid tariff level, which is due to the monthly peak grid tariff. Maximum import is also reduced if the spot price level is 4 or 5 times higher, since there are more hours with high spot price at the same time as high demand. Interest rate and PV size do not affect the maximum import.

4.5. Discussion

With regard to comparing the different storage types in the energy community, our results have shown that the optimisation model favours TES, compared to BESS, due to lower investment costs. When we compare Case EC with Case Industry, we see that the storage sizes are increased, while maximum import and total costs are reduced.

When investigating the different cost distribution methods, we see that only the Shapley value lowers the costs for both the industry consumer and the urban area. We also see that the other cost distribution methods give very different results, and that cost recovery with regard to the investment costs might not occur. This leads to a greater discussion on energy community equity, which might also affect the interest in joining the community. The energy community is assumed to be collaborative, since the optimisation is done centrally, but the cost distribution method strongly affects which member receives the most benefit out of joining the energy community. In this case study, it is the industry consumer that invests in the energy storage. The urban area, on the other hand, contributes with PV generation, which is a reason for why a larger energy storage size is optimal. The cost distribution methods that focus on peak load of the grid clearly do not compensate the urban area, since the PV is not contributing to reducing the peak load effectively in this Norwegian energy community.

We performed a sensitivity analysis to investigate how the results change with different assumptions for input parameters. When we altered the storage investment costs, we found that they highly impacted the optimal size of the storages. It was also more likely that BESS was invested in when optimising for the whole energy community, than for the industry consumer alone. The costs, however, only varied slightly, leading to the conclusion that our results are robust. When we altered the spot price level, grid tariff level, interest rate and PV size, we also found that our results did not change much. There were two exceptions: the costs, which were highly sensitive to the spot price level, and the TES sizes, which were highly sensitive to the grid tariff level. We also saw that the optimal TES size did not follow a linear trend for higher spot price levels and PV sizes. This emphasises the complex nature of optimisation models with many different price-signals, especially with a monthly peak grid tariff. Although the sensitivity analysis showed only minor changes in the optimal TES size for the different spot price levels and PV sizes, the operation of the TES varied depending on whether it was profitable to use the TES for storing surplus PV generation, doing energy arbitrage, or peak shaving.

It should be noted that we have not included a detailed battery degradation model, meaning that the battery may be even less profitable than these results depict. The investment cost for TES has been modelled linearly, which might lead to unrealistically low costs for small sizes of TES. Another point which is important to mention is the space limitations of the energy storages. In this work, we have assumed that the dairy has no space limitations for installing hot water tanks or battery containers. This could certainly play an important role in other case studies where the industry consumer has less space available. In such scenarios, this could potentially bias the outcomes in favour of BESS, given that batteries exhibit a higher density compared to hot water tanks.

This research investigates two perspectives at once: the members of the energy community and the DSO. For the members, the objective of creating the energy community is to lower the costs. This cost reduction, however, occurs because the energy community is assumed to have aggregated net metering. But for the DSO to allow this kind of metering, it needs to be evident that the peak load of the grid is also lowered, so that the energy community also brings benefits to the local grid. The results show that this does in fact happen: When the industry optimises on its own (Case Industry), the maximum peak is reduced to 17.14 MWh/h. When the energy community optimises together (Case EC), the maximum peak is reduced to 16.78 MWh/h. This reduction in maximum import happens since the monthly peak grid tariff is the most important cost driver.

Some results of this study can be compared to existing literature. Refs. [45,46] both concluded that a hybrid solution with BESS and TES was the most economical option. Ref. [45] found that the optimal system was 71% TES and 29% BESS. The authors assumed that the TES had a capital cost 6x lower than the battery, and investigated one commercial building with PV generation and electric vehicle charging. Similarly, [46] found that a hybrid energy storage system with TES and BESS was more cost-effective than single energy storages. The energy storage technologies investigated were different from the ones in this study: a molten salt TES and a lead–acid BESS. Moreover, only electrical demand was included, and therefore no thermal demand. Our finding is in line with [23], which also concluded that it was optimal to invest in TES, and no BESS, for a dairy. The discrepancies between the results are likely related to the thermal share of the demand profile used, where our study and [23] involves an industrial consumer with a significant thermal demand, emphasising the benefit of a TES.

As argued in the introduction, many energy community studies mainly investigate households and disregard industrial consumers. Our findings indicate that industrial consumers constitute a significant part of a local urban area's load, and therefore they are able to make a larger contribution to the local urban area by reducing the peak load. Our findings also indicate that this is in fact profitable for both the industrial consumer and the urban area if an equitable cost distribution method is used. The peak load of this case study is reduced by approximately 0.9 MW, which accounts for 5% of the transformer load. As explained in the introduction, we need to free up as much space in the grid as possible by reducing the peak load. In this particular case, an investment in shared energy storage in an industrial energy community is profitable for the members and contributes to 0.9 MW of new capacity in the grid. As Fig. 5 showed, there are more commercial and industrial consumers in this urban area. If more of these have thermal demand, the peak load might be reduced even further by investing in more or larger TES.

5. Conclusion

In this article, we aimed to quantify the benefits of investing in thermal and electrical energy storage in an industrial energy community, for an industry consumer and the energy community as a whole. We investigated a real-life case study in Trondheim, Norway, using measurements from the local transformer and the industrial consumer.

The results showed that in an industrial energy community with thermal demand, hot water TES was the most favourable storage option, due to lower investment costs than BESS. Furthermore, we found that optimising the storage sizes for the whole energy community leads to both cost reduction for the energy community and a reduction in maximum import for the local grid. The costs were reduced by 1.8%, while the maximum import was reduced by 5%, compared to the reference case. The optimal TES size when optimising for the energy community was 16% higher than the optimal TES size when optimising for the industrial consumer alone. BESS was only economically viable in the sensitivity analysis, for a significant reduction in BESS investment cost, or a significant increase in spot price or grid tariff levels. In this particular case study, an investment in shared energy storage at an industrial energy community is profitable for the actors included, and contributes to 0.9 MW of new capacity in the grid.

When investigating the cost distribution between the industrial consumer and the urban area, we found that the cost distribution method heavily impacts whether it is economically attractive for the industry consumer to join the energy community. Only the coincident peakmethod and the Shapley value lead to a reduction in electricity costs for the industrial consumer. Future work should investigate how to divide these costs fairly, ensuring that members who invest in capital-intensive technologies, such as energy storage, can recover their costs.

The sensitivity analysis showed that BESS investment costs must decline significantly for it to be a competitive option compared to TES. Therefore, in industrial energy communities, these results support that TES should be invested in where possible before considering investing in BESS. The sensitivity analyses also showed that an increase in PV generation leads to a reduction in TES size. These results show that forming energy communities with the inclusion of industry actors with thermal demand may be profitable and should be given attention in future energy community studies.

CRediT authorship contribution statement

Kjersti Berg: Writing – review & editing, Writing – original draft, Visualization, Software, Methodology, Formal analysis, Data curation, Conceptualization. **Sverre Stefanussen Foslie:** Writing – review & editing, Writing – original draft, Validation, Methodology, Investigation, Data curation, Conceptualization. **Hossein Farahmand:** Writing – review & editing, Supervision, Funding acquisition, Conceptualisation.



Fig. A.16. Overview of the electrical and thermal energy flows at the industry consumer investigated in this particular case study. (p) denotes parameters and (v) denotes variables in the optimisation model.

Declaration of competing interest

The authors declare the following financial interests/personal relationships which may be considered as potential competing interests: Kjersti Berg reports administrative support and statistical analysis were provided by Tensio TS. Kjersti Berg reports administrative support and statistical analysis were provided by Tine SA. If there are other authors, they declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

The authors do not have permission to share data.

Acknowledgements

This work was supported by the FINE project [project number 308833], and partly by the projects [323330] and [257632], all funded by the Research Council of Norway. We thank TINE SA for providing us with measurements and information about their system. Furthermore, we thank Tensio TS for providing us with measurements and information regarding the area load. We also thank Marthe F. Dynge for discussions regarding cost distribution.

Appendix A. Constraints for industry consumer

Fig. A.16 shows an overview of the electrical (blue) and thermal (red) systems at the industry consumer site. The 400 V bus supplies electrical demand and the 690 V bus supplies an electric boiler that provides thermal energy. The optimisation model has the option to install a BESS at the 400 V bus and a TES further downstream from the 690 V bus as shown in the figure.

The energy balance constraints for electricity and heat at the industry facility are given in (A.1)–(A.8).

$p_t^{\text{industry}} = p_t^{400\text{V}} + p_t^{690\text{V}}$	$t \in T$ (A.1)
$p_t^{690V} = P_t^{\text{steam boiler}} + p_t^{\text{el. boiler}}$	$t \in T$ (A.2)
$q_t^{\text{el. boiler}} + Q_t^{\text{heat pump}} = q_t^{\text{hot water}} + Q_t^{\text{other}}$	$t \in T$ (A.3)
$p_t^{\text{el. boiler}} \leq \overline{P^{\text{el. boiler}}}$	$t \in T$ (A.4)
$p_t^{\text{el. boiler}} \eta^{\text{el. boiler}} = q_t^{\text{el. boiler}}$	$t \in T$ (A.5)

K. Berg et al.

$$\begin{aligned} q_t^{\text{hot water}} &= Q_t^{\text{process}} + q_t^{\text{TES,ch}} - q_t^{\text{TES,dis}} & t \in T \text{ (A.6)} \\ p_t^{400V} &= P_t^{\text{other}} + p_t^{\text{BESS,ch}} - p_t^{\text{BESS,dis}} & t \in T \text{ (A.7)} \\ p_t^{690V}, p_t^{400V}, p_t^{\text{el. boiler}}, q_t^{\text{el. boiler}}, q_t^{\text{hot water}} \ge 0 & t \in T \text{ (A.8)} \end{aligned}$$

Appendix B. Data handling

The data from the dairy is gathered from the energy monitoring system of the dairy, covering both electric and thermal systems. The measurements are hourly for 2022. As there were some errors in the data monitoring system during some periods, some data had to be approximated. Following is an explanation of the changes.

B.1. Thermal measurements

The data management system logging the thermal side had some errors from January to mid-February of 2022. Due to this, the measurements on the thermal side for this period could not be used. New data for this period was generated, replacing weeks 1 to 7 with week 8. There is a slight temperature dependence in the demand for thermal energy from the boilers and heat pump, however the temperatures in January and February 2022 at the location of the dairy were fairly equal. We therefore consider this approximation acceptable.

Furthermore, in 42 h, the heat pump thermal energy output was higher than the measured thermal demand, leading to an infeasible optimisation problem when running for Case Ref. Therefore, in these hours, the heat pump thermal power output was set equal to the demand.

B.2. Electricity measurements

During some periods, the measurement values for the electricity consumption of the steam boiler were missing. Some of these are single missing values, while the main part are continuous missing values in the periods: April 12–April 20, April 30 (6 h), May 28–June 03, June 26–June 27, August 05–August 22. In total, approx. 12.5% of the values are missing. In these periods, the gap has been filled using either data from the day before, or the weeks before or after. The steam boiler has a small electricity consumption compared to the electric boiler, so any small error here would not affect the overall results significantly.

B.3. Peaks

Some outliers have been identified in the measurements of the thermal energy demand, which are obviously outside the operational range of the system. The measurements which are negative, or above 2.5 MW (other thermal demand) and 2 MW (process demand) have been adjusted to comply with the hour before the misreading. In total, 23 values were modified this way.

The measurements from the transformer station had two days of extraordinary measurements with a very high peak (18 MW) on June 22 and zero consumption on June 23, due to maintenance work. These days have been set equal to the day before and after, June 21 and June 24, respectively.

B.4. Daylight saving time

In the hour of switching from daylight saving time in October, the measurements of the two consecutive hours of 02:00 were merged. To correct this, they were divided into two equal parts in the consecutive hours.

References

- Haque MM, Wolfs P. A review of high PV penetrations in LV distribution networks: Present status, impacts and mitigation measures. Renew Sustain Energy Rev 2016;62:1195–208. http://dx.doi.org/10.1016/j.rser.2016.04.025, URL https://www.sciencedirect.com/science/article/pii/S1364032116300429.
- [2] Hennig RJ, de Vries LJ, Tindemans SH. Risk vs.restriction—An investigation of capacity-limitation based congestion management in electric distribution grids. Energy Policy 2024;186:113976. http://dx.doi.org/10.1016/ j.enpol.2023.113976, URL https://www.sciencedirect.com/science/article/pii/ S030142152300561X.
- [3] Lieberwirth M, Hobbie H. Decarbonizing the industry sector and its effect on electricity transmission grid operation—Implications from a model based analysis for Germany. J Clean Prod 2023;402:136757. http://dx.doi.org/10. 1016/j.jclepro.2023.136757, URL https://linkinghub.elsevier.com/retrieve/pii/ S0959652623009150.
- [4] Statnett. Major investments to ensure the green change of pace. 2023, URL https://www.statnett.no/en/about-statnett/news-and-press-releases/newsarchive-2022/major-investments-to-ensure-the-green-change-of-pace/. [Accessed 08 December 2023].
- [5] Nakstad, et al. Nett i tide [eng.: grid in time]. Tech. rep., 2022, URL https: //www.regjeringen.no/contentassets/9dabbb7fb58e4bb297f4388696570460/no/ pdfs/nou202220220006000dddpdfs.pdf.
- [6] National report 2021. Tech. rep., The Norwegian Energy Regulatory Authority (RME); 2022, URL https://publikasjoner.nve.no/rme_rapport/2022/rme_ rapport2022_08.pdf.
- [7] Golmohamadi H. Demand-side management in industrial sector: A review of heavy industries. Renew Sustain Energy Rev 2022;156:111963. http://dx.doi. org/10.1016/j.rser.2021.111963, URL https://www.sciencedirect.com/science/ article/pii/S1364032121012284.
- [8] Hernandez-Matheus A, Löschenbrand M, Berg K, Fuchs I, Aragüés-Peñalba M, Bullich-Massagué E, Sumper A. A systematic review of machine learning techniques related to local energy communities. Renew Sustain Energy Rev 2022;170:112651. http://dx.doi.org/10.1016/j.rser.2022.112651, URL https:// www.sciencedirect.com/science/article/pii/S1364032122005433.
- [9] Berg K, Rana R, Farahmand H. Quantifying the benefits of shared battery in a DSO-energy community cooperation. Appl Energy 2023;343:121105. http: //dx.doi.org/10.1016/j.apenergy.2023.121105, URL https://www.sciencedirect. com/science/article/pii/S0306261923004695.
- [10] Good N, Mancarella P. Flexibility in multi-energy communities with electrical and thermal storage: A stochastic, robust approach for multi-service demand response. IEEE Trans Smart Grid 2019;10(1):503–13. http://dx.doi.org/10. 1109/TSG.2017.2745559, URL https://ieeexplore.ieee.org/abstract/document/ 8052516. conference Name: IEEE Transactions on Smart Grid.
- [11] Directive (EU) 2018/2001 of the European parliament and of the council of 11 December 2018 on the promotion of the use of energy from renewable sources. The European Commission; 2018, URL https://eur-lex.europa.eu/eli/dir/2018/ 2001/2018-12-21.
- [12] Directive (EU) 2019/944 of the European parliament and of the council of 5 June 2019 on common rules for the internal market for electricity and amending directive 2012/27/ EU. The European Commission; 2019, URL https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=uriserv:OJ.L_ .2019.158.01.0125.01.ENG&toc=OJ:L:2019:158:TOC.
- [13] Petrovich B, Kubli M. Energy communities for companies: Executives' preferences for local and renewable energy procurement. Renew Sustain Energy Rev 2023;184:113506. http://dx.doi.org/10.1016/j.rser.2023.113506, URL https: //www.sciencedirect.com/science/article/pii/S1364032123003635.
- [14] Industry. IEA; 2023, https://www.iea.org/energy-system/industry. [Accessed 28 November 2023].
- [15] Eslamizadeh S, Ghorbani A, Künneke R, Weijnen M. Can industries be parties in collective action? Community energy in an Iranian industrial zone. Energy Res Soc Sci 2020;70:101763. http://dx.doi.org/10.1016/j.erss.2020.101763, URL https://www.sciencedirect.com/science/article/pii/S2214629620303388.
- [16] Barabino E, Fioriti D, Guerrazzi E, Mariuzzo I, Poli D, Raugi M, Razaei E, Schito E, Thomopulos D. Energy communities: A review on trends, energy system modelling, business models, and optimisation objectives. Sustain Energy Grids Netw 2023;36:101187. http://dx.doi.org/10.1016/j.segan.2023.101187, URL https://www.sciencedirect.com/science/article/pii/S2352467723001959.
- [17] Caballero V, Briones A, Coca-Ortegón A, Pérez A, Barrios B, de la Mano M. Analysis and simulation of an urban-industrial sustainable energy community: A use case in San Juan de Mozarrifar using photovoltaic energy. Energy Rep 2023;9:1589–605. http://dx.doi.org/10.1016/j.egyr.2022.12.059, URL https: //www.sciencedirect.com/science/article/pii/S2352484722026609.
- [18] Butturi MA, Lolli F, Sellitto MA, Balugani E, Gamberini R, Rimini B. Renewable energy in eco-industrial parks and urban-industrial symbiosis: A literature review and a conceptual synthesis. Appl Energy 2019;255:113825. http://dx.doi.org/ 10.1016/j.apenergy.2019.113825, URL https://www.sciencedirect.com/science/ article/pii/S0306261919315120.

- [19] Bartolini A, Carducci F, Muñoz CB, Comodi G. Energy storage and multi energy systems in local energy communities with high renewable energy penetration. Renew Energy 2020;159:595–609. http://dx.doi.org/10.1016/ j.renene.2020.05.131, URL https://www.sciencedirect.com/science/article/pii/ S0960148120308351.
- [20] Zhang X, Hug G, Kolter JZ, Harjunkoski I. Demand response of ancillary service from industrial loads coordinated with energy storage. IEEE Trans Power Syst 2018;33(1):951–61. http://dx.doi.org/10.1109/TPWRS.2017.2704524, URL https://ieeexplore.ieee.org/abstract/document/7930447. conference Name: IEEE Transactions on Power Systems.
- [21] Martins R, Hesse HC, Jungbauer J, Vorbuchner T, Musilek P. Optimal component sizing for peak shaving in battery energy storage system for industrial applications. Energies 2018;11(8):2048. http://dx.doi.org/10.3390/ en11082048, URL https://www.mdpi.com/1996-1073/11/8/2048. number: 8 Publisher: Multidisciplinary Digital Publishing Institute.
- [22] Parra D, Walker GS, Gillott M. Are batteries the optimum PV-coupled energy storage for dwellings? Techno-economic comparison with hot water tanks in the UK. Energy Build 2016;116:614–21. http://dx.doi.org/10.1016/ j.enbuild.2016.01.039, URL https://www.sciencedirect.com/science/article/pii/ S037877881630038X.
- [23] Foslie SS, Knudsen BR, Korpås M. Integrated design and operational optimization of energy systems in dairies. Energy 2023;281:128242. http:// dx.doi.org/10.1016/j.energy.2023.128242, URL https://www.sciencedirect.com/ science/article/pii/S0360544223016365.
- [24] Parra D, Swierczynski M, Stroe DI, Norman SA, Abdon A, Worlitschek J, O'Doherty T, Rodrigues L, Gillott M, Zhang X, Bauer C, Patel MK. An interdisciplinary review of energy storage for communities: Challenges and perspectives. Renew Sustain Energy Rev 2017;79:730–49. http://dx.doi.org/10. 1016/j.rser.2017.05.003, URL https://www.sciencedirect.com/science/article/ pii/S1364032117306263.
- [25] Cao Z, Zhang M, Zhai C, Wang Y. Scheduling optimization of shared energy storage station in industrial park based on reputation factor. Energy Build 2023;299:113596. http://dx.doi.org/10.1016/j.enbuild.2023.113596, URL https: //www.sciencedirect.com/science/article/pii/S0378778823008265.
- [26] Zhang Z, Zhou K, Yang S. Optimal selection of energy storage system sharing schemes in industrial parks considering battery degradation. J Energy Storage 2023;57:106215. http://dx.doi.org/10.1016/j.est.2022.106215, URL https: //www.sciencedirect.com/science/article/pii/S2352152X22022046.
- [27] Sæther G, Crespo Del Granado P, Zaferanlouei S. Peer-to-peer electricity trading in an industrial site: Value of buildings flexibility on peak load reduction. Energy Build 2021;236:110737. http://dx.doi.org/10.1016/j.enbuild.2021.110737, URL https://linkinghub.elsevier.com/retrieve/pii/S0378778821000219.
- [28] Weckesser T, Dominković DF, Blomgren EMV, Schledorn A, Madsen H. Renewable energy communities: Optimal sizing and distribution grid impact of photo-voltaics and battery storage. Appl Energy 2021;301:117408. http://dx. doi.org/10.1016/j.apenergy.2021.117408, URL https://www.sciencedirect.com/ science/article/pii/S0306261921008059.
- [29] Li N, Hakvoort RA, Lukszo Z. Cost allocation in integrated community energy systems - A review. Renew Sustain Energy Rev 2021;144:111001. http: //dx.doi.org/10.1016/j.rser.2021.111001, URL https://www.sciencedirect.com/ science/article/pii/S1364032121002914.
- [30] Cremers S, Robu V, Zhang P, Andoni M, Norbu S, Flynn D. Efficient methods for approximating the Shapley value for asset sharing in energy communities. Appl Energy 2023;331:120328. http://dx.doi.org/10.1016/j.apenergy.2022. 120328. URL https://linkinghub.elsevier.com/retrieve/pii/S0306261922015859.
- [31] Zabihinia Gerdroodbari Y, Khorasany M, Razzaghi R, Heidari R. Management of prosumers using dynamic export limits and shared community energy storage. Appl Energy 2024;355:122222. http://dx.doi.org/10.1016/j. apenergy.2023.122222, URL https://www.sciencedirect.com/science/article/pii/ S0306261923015866.

- [32] Li L, Peng K, Yang X, Liu K. Coordinated design of multi-stakeholder community energy systems and shared energy storage under uncertain supply and demand: A game theoretical approach. Sustainable Cities Soc 2024;100:105028. http: //dx.doi.org/10.1016/j.scs.2023.105028, URL https://www.sciencedirect.com/ science/article/pii/S221067072300639X.
- [33] Gomes L, Vale Z. Costless renewable energy distribution model based on cooperative game theory for energy communities considering its members' active contributions. Sustainable Cities Soc 2024;101:105060. http://dx.doi.org/ 10.1016/j.scs.2023.105060, URL https://www.sciencedirect.com/science/article/ pii/S2210670723006704.
- [34] Ahrens MU, Foslie SS, Moen OM, Bantle M, Eikevik TM. Integrated high temperature heat pumps and thermal storage tanks for combined heating and cooling in the industry. Appl Therm Eng 2021;189:116731. http://dx.doi.org/ 10.1016/j.applthermaleng.2021.116731, publisher: Pergamon.
- [35] Dalen HM, Halvorsen B, Larsen BM. Strømproduksjon fra solcelleanlegg i norske husholdninger. Analyser av plusskunder basert påelhub. Tech. rep. SSB, 2022.
- [36] Pfenninger S, Staffell I. Long-term patterns of European PV output using 30 years of validated hourly reanalysis and satellite data. Energy 2016;114:1251–65. http://dx.doi.org/10.1016/j.energy.2016.08.060, URL https: //www.sciencedirect.com/science/article/pii/S0360544216311744.
- [37] Tensio TS. Nettleiepriser for bedrift gjeldende fra 1. 2023, juli 2022 [eng.: Grid tariff prices for companies as of 1 July 2022], https://ts.tensio.no/kunde/ nettleie-priser-og-avtaler/2022-nettleie-bedrift-01.07. [Accessed 28 November 2023].
- [38] Campos Celador A, Odriozola M, Sala JM. Implications of the modelling of stratified hot water storage tanks in the simulation of CHP plants. Energy Convers Manage 2011;52(8):3018–26. http://dx.doi.org/10.1016/ j.enconman.2011.04.015, URL https://www.sciencedirect.com/science/article/ pii/S0196890411001452.
- [39] Li H, Hou J, Hong T, Ding Y, Nord N. Energy, economic, and environmental analysis of integration of thermal energy storage into district heating systems using waste heat from data centres. Energy 2021;219:119582. http:// dx.doi.org/10.1016/j.energy.2020.119582, URL https://www.sciencedirect.com/ science/article/pii/S036054422032689X.
- [40] Innovation outlook: Thermal energy storage. IRENA; 2020, URL https://www. irena.org/publications/2020/Nov/Innovation-outlook-Thermal-energy-storage.
- [41] Electricity storage and renewables: Costs and markets to 2030. Tech. rep., IRENA; 2017, URL https://www.irena.org/-/media/Files/IRENA/Agency/ Publication/2017/Oct/IRENA_Electricity_Storage_Costs_2017.pdf.
- [42] Technology data for energy storage. Danish Energy Agency; 2024, URL https://ens.dk/en/our-services/projections-and-models/technologydata/technology-data-energy-storage. [Accessed 25 March 2024].
- [43] Hart WE, Watson J-P, Woodruff DL. Pyomo: modeling and solving mathematical programs in python. Math Program Comput 2011;3(3):219–60.
- [44] Gurobi optimizer reference manual. Gurobi Optimization, LLC; 2023, URL https: //www.gurobi.com. [Accessed 28 November 2023].
- [45] Brandt M, Woods J, Tabares-Velasco PC. An analytical method for identifying synergies between behind-the-meter battery and thermal energy storage. J Energy Storage 2022;50:104216. http://dx.doi.org/10.1016/j.est.2022.104216, URL https://www.sciencedirect.com/science/article/pii/S2352152X2200247X.
- [46] He Y, Guo S, Zhou J, Song G, Kurban A, Wang H. The multi-stage framework for optimal sizing and operation of hybrid electrical-thermal energy storage system. Energy 2022;245:123248. http://dx.doi.org/10.1016/j.energy.2022.123248, URL https://www.sciencedirect.com/science/article/pii/S0360544222001517.